

A Service of

ZBU

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Commins, Nicola; Nolan, Anne

Working Paper Car Ownership and Mode of Transport to Work in Ireland

ESRI Working Paper, No. 310

Provided in Cooperation with: The Economic and Social Research Institute (ESRI), Dublin

Suggested Citation: Commins, Nicola; Nolan, Anne (2009) : Car Ownership and Mode of Transport to Work in Ireland, ESRI Working Paper, No. 310, The Economic and Social Research Institute (ESRI), Dublin

This Version is available at: https://hdl.handle.net/10419/50167

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



Working Paper No. 310

August 2009

Car Ownership and Mode of Transport to Work in Ireland¹

Nicola Commins and Anne Nolan

Abstract: Rapid economic and demographic change in Ireland over the last decade, with associated increases in car dependence and congestion, has focused policy on encouraging more sustainable forms of travel. In this context, knowledge of current travel patterns and their determinants is crucial. In this paper, we extend earlier Irish research to examine the joint decision of car ownership and mode of transport to work. We employ cross-section micro-data from the 2006 Census of Population to estimate discrete choice models of car ownership and commuting mode choice for four sub-samples of the Irish population, based on residential location. Empirical results suggest that travel and supply-side characteristics such as travel time, costs, work location and public transport availability, as well as demographic and socio-economic characteristics such as age and household composition have significant effects on these decisions.

Corresponding Author: anne.nolan@esri.ie

JEL Codes: R41, C51, D12

Key words: Mode Choice; Car Ownership; Conditional Logit; Ireland

¹ The authors would like to thank ESRI seminar participants and participants at the 4th Kuhmo-Nectar Conference on Transport Economics 2009 for helpful comments on an earlier draft.

ESRI working papers represent un-refereed work-in-progress by members who are solely responsible for the content and any views expressed therein. Any comments on these papers will be welcome and should be sent to the author(s) by email. Papers may be downloaded for personal use only.

Car Ownership and Mode of Transport to Work in Ireland

1 Introduction

As a result of rapid economic and demographic change over the last decade, and the resulting increase in car ownership, Ireland has experienced many of the problems associated with increasing car dependence. Over the period 1996-2006², the population of Ireland grew by 16.9 per cent while the numbers in employment increased by 47.6 per cent, largely due to increases in the rate of female participation in the labour force and inward migration. In terms of the implications for transport, the most striking is the increase in new vehicle registrations, which increased by over 60 per cent over the period (Central Statistics Office, 2007). Data for journeys to work, school and college confirm this shift towards the private car; the proportions driving to work increased from 46.3 per cent in 1996 to 57.1 per cent in 2006 (see Figure 1), while the proportion of primary school students travelling as car passengers increased from 35.8 per cent in 1996 to 55.0 per cent in 2006, overtaking the proportions walking (24.3 per cent), which has traditionally been the primary means of transport to school for this age-group. The resulting levels of congestion impact on all those using the road and public transport network; in the Dublin area, average journey speeds in the morning peak for car and bus³ decreased by 12.4 per cent and 6.2 per cent respectively between 2003 and 2004 (Dublin Transportation Office, 2005). There are also wider economic impacts, with carbon dioxide emissions from transport increasing by 88.7 per cent between 1996 and 2006 (Lyons et al., 2008).

[insert Figure 1 here]

Environmental considerations imply a need to reverse or at the very least to halt this shift in favour of the private car. Current policy focuses on a variety of measures that seek to limit or redirect travel demand in the short- to medium-term and encourage alternative more sustainable

² Economic activity has contracted sharply since late 2007. Unemployment reached 11.4 per cent in February 2009 (Central Statistics Office, 2009a), a return to net emigration is forecast for 2009 and 2010 (Barrett *et al.*, 2009) and new car registrations fell by 63.6 per cent between March 2008 and March 2009 (Central Statistics Office, 2009b). ³ Pue areada an Ovelity Pue corridor metric (that is neutron with dedicated read for hyper) and

³ Bus speeds on Quality Bus corridor routes (that is, routes with dedicated road space for buses) only.

land-use strategies in the longer term (see Department of Transport, 2008a, 2008b, Dublin Transportation Office, 2001, 2006a, 2006b, European Commission, 2007, FitzGerald et al., 2008, Morgenroth and FitzGerald, 2006). Investment in public transport and measures which seek to use existing infrastructure more efficiently such as improved cycle and bus lanes, parking restrictions, road pricing, carpooling *etc.* are all considered necessary if a shift away from the private car towards more sustainable methods of transport such as walking, cycling and public transport is to be achieved. Current initiatives include the provision of tax relief for the purchase of public transport tickets and bicycles for commuting trips with more severe measures such as urban road pricing or the introduction of a carbon tax proposed but yet to be implemented.

In this context, knowledge of the factors influencing the demand for passenger transport is crucial. In this paper we concentrate on transport demand for a specific journey purpose, namely the journey to work, and examine the influence of demographic, socio-economic and supply-side factors on choice of mode of transport for the journey to work in Ireland in 2006 using discrete choice econometric methodologies. We extend previous Irish research to incorporate the endogeneity of the car ownership decision by estimating a joint model of car ownership and mode of transport to work. The 2006 Census of Population also contains detailed information on home and work location for the full population of working individuals, allowing us to consider the influence of proximity to rail connections for the first time. Section 2 discusses previous literature in the area, both international and Irish. Section 3 describes the data and provides some descriptive statistics, while Section 4 describes the econometric methodology employed. Section 5 presents empirical results and Section 6 concludes.

2 Previous Research

Internationally, there is an extensive research literature on the determinants of various aspects of travel behaviour, and in particular commuting behaviour. Due to the nature of such decisions, and the data available, discrete or qualitative choice methods such as multinomial or conditional

logit⁴ are typically employed. The models are grounded in consumer utility theory whereby the individual chooses among alternatives with the aim of maximising personal utility. Ben-Akiva and Lerman, 1975 apply the multinomial logit methodology to the choice between a number of different alternatives for the journey to work in Washington, and find particularly significant effects for lifecycle and public transport availability. Aside from modal choice, the multinomial logit methodology has been extensively applied to other transport decisions such as the number of cars to own (Alperovich et al., 1999, Bhat and Pulugurtha, 1998 and Cragg and Uhler, 1970), choice of car type (Lave and Train, 1979 and McCarthy, 1996), tourist destination (Eymann and Ronning, 1997) and choice of departure time (McCafferty and Hall, 1982). A number of studies have analysed mode choice for other journey purposes, using a variety of methods (see Cohen and Harris, 1998 for trips to visit friends and relatives, Domencich and McFadden, 1975 for shopping trips, Ewing et al., 2004 for mode choice for the journey to school and McGillivray, 1972 for other journey purposes including personal business, visiting friends and relations, shopping and other recreation).

Asensio, 2002, De Palma and Rochat, 2000, Dissanayake and Morikawa, 2005, Thobani, 1984 and Train, 1980 all use the nested multinomial logit methodology to estimate modal choice for the journeys to work in Barcelona, Geneva, Bangkok, Karachi and San Francisco respectively. The nested multinomial logit model overcomes the restrictive requirement of the multinomial logit methodology to have distinct and independent alternatives. More recent versions of the nested multinomial logit model (such as the generalised or cross-nested logit) have been developed to incorporate situations in which correlations exist between alternatives across nests as well as alternatives within nests, thus allowing for the incorporation of related decisions such as car ownership or residential/employment location. Vega and Reynolds-Feighan, 2008 estimate a joint model of residential location and mode of transport to work for those working in four areas of Dublin, while Salon, 2009 estimates a joint residential location-car ownership-mode of

⁴ The multinomial logit and conditional logit models differ in the type of explanatory variables that can be included; the conditional model can support individual-specific as well as alternative-specific variables while the multinomial logit can support only the former.

transport to work model for residents of New York City (although model selection tests favour the conditional logit).⁵

Much of the early research on Irish travel patterns was carried out in the context of research on the sustainability of residential and commercial development (see for example, MacLaran and Killen, 2002, McCarthy, 2004 and Williams and Shiels, 2000). The interactions between commuting and the housing and labour markets have been analysed by Morgenroth, 2002 who used gravity models to analyse the determinants of inter-county commuting flows and Keane, 2001 who similarly related commuting to issues of job search and the development of local labour market areas. Horner, 1999 and Walsh *et al.*, 2005 described patterns of travel to work using earlier versions of the Census of Population (COP) data employed in this paper. Both papers highlighted a substantial phenomenon of long-distance commuting.

Research on the travel behaviour of individuals using disaggregated data has been increasing in recent years in Ireland, in part due to the increased availability of detailed micro-data on commuting behaviour from the Census of Population. Nolan, 2003 examined the income and socio-economic determinants of household car ownership, car use and public transport use in the Dublin area, using micro-data from the 1987, 1994 and 1999 Irish Household Budget Surveys. McDonnell et al., 2006 focused on the determinants of bus use in a particular QBC (quality bus corridor) catchment area in Dublin. They found that the key to attracting commuters to bus was shorter journey times at peak times, even in high income areas. Vega and Reynolds-Feighan, 2006 estimated a simultaneous model of residential location and mode of transport to work in the Dublin area using data from the 2002 Census of Population, and found significant effects for alternative-specific characteristics such as travel time, as well as individual socio-economic characteristics. In a later paper, using the same data, Vega and Reynolds-Feighan, 2008 concentrated on four employment sub-centres in the Dublin area, and found that the spatial distribution of employment exerted a large and significant influence on modal choice for the

⁵ De Donnea, 1971, Lave, 1970 and Madan and Groenhout, 1987 all use the binary logit methodology but the ability of the conditional, multinomial and nested logit methods to incorporate more than two categories of the dependent variable means that they are favoured in applied work relating to modal choice. Bhat and Pulugurtha, 1998 and Hausman and Wise, 1978 estimate multinomial probit models, but the computational complexity of this model means that it is rarely applied.

journey to work. Commins and Nolan, 2008, using the same data employed in this paper (i.e., the 2006 Census of Population), examined choice of mode of transport for the journey to work in the Greater Dublin Area, but assumed that residential location and household car ownership status were exogenous.

3 Data

The data employed in this paper are micro-data from the Place of Work Census of Anonymised Records (POWCAR) from the 2006 Census of Population (COP). The COP is carried out every five years by the Central Statistics Office and includes all individuals present in the country on the last Sunday in April. For the first time, the micro-data for 2006 constitute the entire population of working individuals aged 15+ years surveyed at home in private households. In total 1,834,472 individuals are included in the micro-data file. After excluding individuals working from home, those with a mobile place of employment and where "other means"⁶ and lorry/van were recorded, the final sample for estimation is 1,564,330 individuals. Due to the substantial difference in population density and public transport provision across different areas of Ireland, we further divide the sample into four sub-samples; Dublin city and county (494,370 individuals), Dublin commuter belt (i.e., the surrounding counties of Kildare, Meath and Wicklow; 187,779 individuals), other urban areas (377,649 individuals) and rural areas (504,532 individuals).⁷ Table 1 defines the four sub-samples, and provides some details on public transport availability and transport characteristics in each area.

[insert Table 1 here]

Each individual observation contains information on demographic and socio-economic characteristics such as age, gender, household type, housing tenure, marital status, education level, socio-economic group and industrial group, as well as variables relating to county and

⁶ These observations are excluded as the modelling approach requires that alternatives be distinct and independent.

⁷ To ease the computational burden, we take a 10 per cent random sample in each case.

electoral division (ED⁸) of residence, county, ED and geo-code of place of work, distance travelled, time of departure and mode of transport for the journey to work. Mode of transport refers to the usual mode of transport for the outward journey to work. Where more than one mode of transport is used, the mode of transport used for the greater part of the journey (by distance) is recorded. Household car ownership refers to the number of cars or vans available for use by the household. All variables are self-reported. The COP does not contain information on income or prices.

Our joint model of household car ownership and mode choice for the journey to work consists of six alternatives; two car ownership levels (no car or at least one car) and three modes of transport to work (walk/cycle, bus/train and motorcycle/car driver/car passenger). See Section 4 for further details on methodology. Table 2 presents car ownership and modal shares for 2006, and indicates that the majority of workers travelled by car in each of the four areas, followed by walking/cycling and public transport. However, it is clear that the range of options available to those in the Greater Dublin Area (i.e., Dublin city and county and commuter belt) is wider, with public transport really only attracting a significant number of commuters here. The proportion of households with at least one car is considerably higher in rural areas than in Dublin city and county. Consequently, the distribution of individuals across all six alternatives is more dispersed for Dublin city and county than for the other areas, in particular, rural areas.

[insert Table 2 here]

Independent variables are individual- as well as alternative-specific. While (self-reported) travel times for the individual's chosen mode are available in POWCAR, travel times for alternative modes are not. To estimate travel times for the non-chosen modes, we regress time on distance for each mode, and use the fitted values to predict average travel times for each of the alternatives for each individual (see De Palma and Rochat, 2000 and Hole and FitzRoy, 2004 for similar applications). Cost information is not available in POWCAR. We construct a simple alternative-specific (monetary) cost per kilometre variable using information on public transport

 $^{^{8}}$ The electoral division (ED) is the smallest administrative area for which population statistics are published. There are 3,440 EDs in the state.

fares and car operating costs (including fuel). We assume zero costs for the walking and cycling modes (in common with others in the literature (see also Hole and FitzRoy, 2004)).⁹

Individual-specific independent variables include the age of the individual (classified using a nine-category variable representing five-yearly age groups) and gender (with males regarded as the reference category). We also include a seven-category household composition variable to identify households with children, single parent households, other households *etc*. This is important as POWCAR does not include household identifiers, meaning that we cannot link household members. Individuals that are married¹⁰ are indicated by a binary variable for marital status, as are individuals with third level education as their highest level of education completed. The socio-economic group of the individual is represented by a four-category variable that identifies individuals in the three highest socio-economic groups (employers and managers, higher professional, lower professional), with those in all other socio-economic groups regarded as the reference category. We include a four-category indicator for industrial group, in an attempt to proxy job characteristics such as flexibility in working hours, provision of company vehicles *etc*. Individuals working in the commercial sector, in public administration and defence, and in education, health and social services are included, with those in all other industrial groups regarded as the reference category.

We also include dummy variables for those living and working in densely populated EDs (i.e., with 150 persons or more per square km). This provides a crude proxy for public transport availability and parking provision with the expectation that those living and working in densely populated areas will have better public transport options and/or poorer parking availability than those living and working in less densely populated areas. We also construct a rail availability index based on ED-level data. This is a binary variable, which identifies individuals who live and work in EDs with 75 per cent of addresses within two kilometres of a rail station (for the Dublin city and county and commuter samples, the cut-off is 100 per cent due to the smaller size of the

⁹ Further details on the construction of the time and cost variables are available from the authors.

¹⁰ Co-habitation is not recorded in the Census.

EDs). Potentially important omitted variables include cycle lane facilities¹¹, bus service availability and more general indicators of public transport quality and frequency, although even if available, matching such variables to EDs would be difficult. Variable definitions and summary statistics are presented in Table 3.

[insert Table 3 here]

4 Methodology

In this application, an individual chooses among six discrete alternatives (representing two car ownership alternatives and three mode of transport alternatives). We specify a conditional logit model, a particular type of discrete choice econometric method. The conditional logit model extends the multinomial logit model to include variables that describe the attributes of the choices (such as travel time), as well as variables that describe the attributes of the individuals (such as age or gender). Assume each individual *i* faces a choice between a set of *J* alternatives (j = 1, 2, ..., J), with the attributes of the choices described by z_{ij} and the characteristics of the individual described by x_i . The model is based on McFadden's random utility framework (see McFadden, 1974), in which each individual *i* aims to maximize their utility. The (unobserved) utility of each alternative is assumed to be a linear function of various independent variables and an error term as follows:

$$U_{ij}^* = x_i'\alpha_j + z_{ij}\beta + \varepsilon_{ij}$$
(1)

where U_{ij}^* is the unobserved utility individual *i* derives from alternative *j*, x_i is the vector of individual-specific independent variables, α_j is the vector of estimated parameters for the individual-specific variables, z_{ij} is the vector of alternative-specific variables, β is the vector of

¹¹ See Ewing *et al.*, 2004 for a discussion of the effect of footpaths and cycle lanes on choice of mode of transport to school in Florida.

alternative-specific parameters and ε_{ij} is the error term. An individual *i* chooses alternative *j* if it gives the highest utility among all possible alternatives. The distributional assumptions concerning the random error component ε_{ij} determine the form of the model. The most common assumption is that the error terms are independently and identically distributed with a Type 1 Extreme Value (or Weibull) distribution, which results in the following probability of individual *i* choosing alternative *j*:

$$\Pr\left(y_{i}=j\right) = \frac{\exp\left(x_{i}\alpha_{j}+z_{ij}\beta\right)}{\sum_{k=1}^{K}\exp\left(x_{i}\alpha_{k}+z_{ik}\beta\right)}$$
(2)

Conditional logit regression methods (using the 'asclogit' command in STATA 10) are used to obtain estimates of the parameters α_j and β . The conditional logit model reduces to the multinomial logit model when all independent variables are individual-specific. As with the multinomial logit, a restrictive feature of the conditional logit model is the assumption of 'Independence from Irrelevant Alternatives' (IIA). The property implies that the relative probabilities between a pair of alternatives are specified without reference to the nature of the other alternatives in the choice set. Hausman and Small-Hsiao tests of the IIA property have been developed for the multinomial logit and conditional logit models, but are prone to errors (see for example, Scott Long and Freese, 2006). To test the appropriateness of the conditional logit model. The assumption of independent alternatives cannot be rejected for the Dublin city and county sample, and while rejected for the other three samples, the inclusive values are either outside the normal range for some nests (indicating inconsistency with utility maximization) and/or the empirical results are very similar in sign and significance to those from the conditional logit model.¹² We therefore concentrate on the estimation of conditional logit models of car ownership and mode choice.

¹² Results are available on request from the authors.

In order to estimate the models, the data must be constructed in such a way that there are J observations for each individual i. As there are 35,528, 13,896, 26,899 and 35,292 individuals in our sample with complete information on all variables of interest respectively, this results in respective sample sizes of 213,168, 83,376, 161,394 and 211,752. Estimation results are presented in terms of odds ratios, with values greater than unity indicating an increased probability of observing the alternative in question, and values smaller than unity a reduced probability of observing the alternative in question (in comparison with the base alternative).

It is possible that each individual does not have access to the full range of alternatives, particularly in rural areas where public transport options may just not be available. We therefore estimate a second specification of the model with a restricted choice set. We consider walking and cycling to be unavailable for those travelling over ten kilometres to work and public transport to be unavailable for those living in EDs with fewer than 100 per cent of addresses within two kilometres of a rail station (see also Ewing et al., 2004 and Hole and FitzRoy, 2004).¹³ As very few individuals who travel by motorised means to work (motorcycle or car) live in households without a car, we also consider the case when this alternative is dropped from the model.¹⁴ Reference to these results is made in Section 5.

5 Empirical Results

Tables 4, 5, 6 and 7 present estimation results for the conditional logit models of car ownership and mode choice for each of the four sub-samples, while Tables 8 and 9 present the time and cost elasticity estimates respectively. Our travel cost variable is necessarily a crude approximation of the monetary costs associated with the various transport modes, but nonetheless, our results indicate that travel cost exerts a negative and significant effect for residents of Dublin city and county, other urban areas and rural areas (as expected). The effect of travel cost is insignificant for residents of the commuter counties. Travel time is a much more significant influence overall

¹³ In the absence of more detailed information on public transport availability, access to rail services at the ED level is used here to proxy, albeit imperfectly, public transport availability.

¹⁴ Results from these various robustness checks are available from the authors.

however; with the effects for the no car-walk or cycle and car-walk or cycle alternatives being particularly pronounced. The cross elasticities of travel time are highest for the car-motorised means alternatives, suggesting that an increase in travel time for this alternative is associated with proportionately large increases in the probability of other alternatives being chosen (e.g., in the commuter belt around Dublin, an increase of one per cent in travel time for those owning cars and choosing motorised means to work leads to a decline of 1.3 per cent in the probability of choosing that alternative, and a 5.2 per cent increase in the probability of the other alternatives).

[insert Tables 8 and 9 here]

The results for the individual-specific variables for Dublin city and county (Table 4), suggest that age has a significant influence on individuals' car ownership and mode choice decisions, with older age groups being significantly less likely to choose all car ownership-mode alternatives in comparison with the base alternative of owning a car and travelling by motorised means (motorcycle, car passenger or car driver) to work. Compared with the base alternative, females are significantly less likely to choose the no car-walk or cycle, no car-public transport and carwalk or cycle options. However, females in Dublin city and county are significantly more likely to choose the car-public transport option than males, perhaps reflecting competing demands on the household car which favour males and the significantly lower probability of females cycling to work which has been observed in other studies (see Commins and Nolan, 2008 and Pooley and Turnbull, 2000). Household composition also proves to be a significant determinant of car ownership and transport mode choice for the journey to work. Households with children are less likely to choose any of the no-car-owning alternatives, compared with single adult households. However, contrary to prior expectations, all other households are more likely to own a car but to walk, cycle or take public transport, than own a car and take motorised means to work, compared with single person households. This may suggest that car ownership is of more importance for non-work trips, particularly when there are children in the household. It may also reflect the fact that our measure of car ownership refers to the number of cars or vans available for use by the entire household, rather than the individual commuter; as such, individuals in larger households face competition for the household car for the journey to work. Married individuals are significantly less likely to choose the public transport alternatives than the base alternative.

While higher education levels are negatively associated with the no-car-owning alternatives, those with a third level education in Dublin also have an increased probability of walking or cycling to work, or opting for public transport, despite owning a car. These divergent effects may suggest that the income effects associated with higher education, which are observed through the greater probability of car ownership, are counteracted by a greater awareness of the detrimental environmental effects of car driving among the higher educated, whose choose more environmentally friendly modes of transport for commuting purposes. Socio-economic group, used as a proxy for household resources, is similarly significant. Those in the top three socioeconomic groups are less likely to choose any alternative over owning a car and driving or travelling as a passenger in a car to work, with the effects for the top two socio-economic groups particularly large and significant, as expected. The results for industrial group suggest that those working in commerce or public administration and defence in the Dublin sample are significantly more likely to choose the two car-owning alternatives. Despite a recent survey which highlighted the high degree of free car parking available to public servants (i.e., those working in public administration) in the Dublin area¹⁵, other characteristics of these occupations such as the availability of subsidised public transport fares and/or their more regular working hours may make them more amenable to non-motorised modes. Those in the education and health industries do not behave significantly differently from those in all other industries (the base category), except that they are significantly less likely to choose the public transport options, perhaps because of irregular/non-peak working hours.

Public transport availability is evidently an important consideration, as shown by the highly significant rail proximity variable. Those living and working in parts of Dublin city and county which are well serviced by rail are significantly more likely to choose all car ownership-mode combinations other than owning a car and travelling by motorised means, with the results for those choosing the no car-walking, cycling and public transport alternatives particularly significant. For instance, in comparison with living in a car-owning household and choosing a

¹⁵ The survey by the Dublin City Business Association suggested that up to 60 per cent of car parking spaces in Dublin city centre were used by public servants, the majority of whom have free parking (Irish Times, June 16th, 2008).

motorised means of transport to work, those living and working in areas with all addresses within 2km of a railway station are nearly three times as likely to choose the no car-public transport option, and over 3.5 times more likely to choose the no car-walking or cycle option. Public transport alternatives, along with walking and cycling, even for those in car-owning households, also become more attractive for those working in densely populated areas.

[insert Table 4 here]

Results for the sample of those living in the commuter belt counties of Kildare, Meath and Wicklow (see Table 5) are broadly in line with those observed for Dublin city and county, with age, household composition, living and working in densely populated areas, and access to rail having the highest significance. These variables all have the same signs as previously outlined, with younger people, those working in densely populated areas and those living and working near a railway station significantly more likely to choose all car ownership-mode combinations over owning a car and travelling by motorised means to work. Individuals in higher socio-economic groups are also still less likely to choose any alternative over owning a car and taking a motorised mode to work, with the effects for choosing the no car-walking or cycling alternative particularly small and significant. Due to the smaller number of observations in this sample, the significance levels of some variables such as gender, marital status and industrial group fall.

[insert Table 5 here]

The results for other urban and rural areas (Tables 6 and 7 respectively) differ in some respects to the samples outlined above, partly due to smaller sample sizes and reduced significance levels. However, the main drivers of car ownership levels and transport mode choice are still clearly evident, in the significance of age, gender, household composition and socio-economic group. The results for the rail availability and work population density variables indicate some differences in comparison with the results for Dublin city and county and the commuter counties. For example, rail availability is less significant for the other urban and rural samples, reflecting the relatively poor availability of rail connections suitable for commuting outside the Greater Dublin Area, although it has, in general, the expected effects. While those working in densely

populated areas are significantly more likely to choose the car-public transport option in both urban and rural areas, those working in densely populated areas are significantly less likely to choose the walking or cycling alternatives in the rural sub-sample. Why this is the case is not entirely clear, although it is possible that conditions for walking and cycling in rural areas are so poor (no cycle lanes, poor lighting, lack of a continuous footpath *etc.*) as to make these alternatives unpopular even for those working in densely populated areas.

[insert Tables 6 and 7 here]

Comparing the results across the four sub-samples indicates differences in the effects of some variables, most notably for education, industrial group and work location population density. While in all cases the probability of choosing a no car alternative is lower among those with a third level education, the possible preference for more environmentally friendly modes of transport among higher educated car owners in Dublin city and county and the commuter counties is not reflected in other parts of the country (where the car-walking or cycling and carpublic transport alternatives are significantly less likely to be chosen). Explaining these divergent effects is difficult although it is possible that education in the Dublin city and county or commuter samples may be correlated with other factors such as work or home location, which may not have been picked up by the aggregated nature of the dummy variables. The effect of workplace population density also differs across the four areas. In general, those resident in rural areas behave differently to those resident in the other three areas of the country; for example, while those working in areas with high population density are significantly more likely to choose the car-walk or cycle option in Dublin city and county, the commuter counties and other urban areas, rural residents who work in densely populated areas are significantly less likely to choose this option. The relative quality of walking and cycling facilities in rural areas may explain this counterintuitive result.

Some of our models are fitting better than others, in part due to amount of variation across alternatives and the quality of independent variables available to us. For example, the Dublin city and county sample has the highest significance levels for all variables, followed by the commuter counties. These areas are better serviced by public transport, and have lower car ownership levels, meaning that individuals are more likely to choose from a wider variety of alternatives than the other urban and rural areas. It also proved more difficult to construct reliable cost variables for areas outside of Dublin, due to insufficient data, which may account for the insignificance of the cost variable for the commuter sample. Finally, the small proportion of individuals choosing the no car-motorised means option leads to less significant results, although consistent with expectations (e.g., non-single households are significantly less likely to choose the no car-motorised option, perhaps reflecting competing household demands such as the presence of school-age children which would make car-sharing more attractive).

Nonetheless, the results from all four samples give a clear indication that the dominant socioeconomic influences on car ownership and modal choice are age, gender, household composition and socio-economic group, regardless of household location. Regional characteristics, such as work and home location are also significant. The strong effects of travel time, along with the positive results for rail availability, suggest some scope for policy intervention. The impact of rail proximity on the probability of travel by public transport suggests that there would be positive effects from the provision of better quality public transport. It would be interesting to test the relative popularity of bus and rail services, given international evidence on potential for rail services (which are not subject to congestion in the same way as bus services) to divert significant amounts of commuters away from motorised means (see for example, Webster and Bly, 1980). A change in the relative time or monetary costs of public transport relative to driving a car are also likely to make these more sustainable modes more attractive to commuters, with our results suggesting that changes to travel time would have stronger effects. Faster or cheaper public transport journeys, or relatively more expensive car trips, perhaps through the introduction of road pricing or carbon taxes have the potential to induce modal shifts. However, influencing behaviour that is associated with individual or household characteristics is more challenging.

We also estimated alternative models for each of the four sub-samples, based on a restricted choice set. In rural areas in particular, it is questionable whether all alternatives are available to all commuters. To restrict the choice set, we exclude walking and cycling as an alternative where the individual travels more than ten kilometres to work, and exclude public transport as an

alternative where the individual lives in an area with poor rail availability. The results¹⁶ are largely consistent with those from the unrestricted models, although significance levels fall.

Finally, we also estimated a simple joint residential location-car ownership-mode choice model for the Dublin city and county sample¹⁷. Our analysis thus far has assumed that residential location is exogenous, but of course, an individuals' mode of transport to work (or preferred mode of transport to work) may also influence their residential location.¹⁸ We estimated a model with twelve alternatives, comprising two residential location (Dublin city centre and not), two car ownership levels (no car and one or more cars) and three modes of transport (walk/cycle, public transport and motorised means). Most of the variables have similar effects on the joint residential location-car ownership-mode decision, with age, gender, household composition, socio-economic group, rail availability and working densely populated areas having particularly consistent effects. The divergent effects of education found for the joint car ownership-mode choice model, where those with a third level education are significantly more likely to choose the car-walk or cycle and car-public transport options, is replicated here, where those with a third level education on the car-public transport option (with no significant difference in the effect of education on the car-public transport option however).

6 Summary and Conclusions

Despite the limitations associated with using Census of Population data to examine modal choice decisions (see Section 3), the results highlight the importance of individual demographic and socio-economic characteristics, travel times and costs, as well as regional and travel variables such as rail availability and home and work location in explaining the joint car ownership-mode of transport decision in Ireland.

¹⁶ Results are available on request from the authors.

¹⁷ Results are available on request from the authors.

¹⁸ Of course, it is also possible that work location is not exogenous either, although as individuals have less choice over their work location, we leave the estimation of such a four-way model as future research.

In the Greater Dublin Area, those working in densely populated areas are significantly more likely to choose the car-walk or cycle and car-public transport options, indicating the effect of public transport availability and city centre parking difficulties and restrictions. The significant positive results observed for public transport use by those working in densely populated areas also add weight to the argument for the development of more concentrated employment districts, to reverse the trends of employment suburbanisation and urban sprawl, which increase car dependence. In addition, in comparison with those with poor rail availability, those living and working in EDs with good rail facilities are in general significantly more likely to choose all options in favour of owning a car and travelling by motorised means to work. This reflects the importance of public transport provision in influencing modal choice. In other urban and rural areas, rail availability and home and work location exert less significant effects, although the results are similar in effect and indicate the importance of public transport provision and parking restrictions on modal choice decisions.

The importance of household or family interactions in determining travel behaviour is confirmed by the significance of age, gender, household type and marital status. Non-single households are significantly less likely to choose the no car options, while females are also significantly less likely to choose these options (but are significantly more likely to choose the car-public transport option). Individual modal choice decisions are often made with reference to other members of the household, in particular with regard to the needs and schedules of school-age children and/or the availability of the household car. In recent years, the proportion of schoolchildren being driven to school has increased substantially, and while the results here are static, the results for household type and marital status to some extent reflect this situation with individuals in households with young children in particular being significantly less likely to choose the no car options.

Of course, the research is subject to a number of caveats. We assume that residential location (and indeed work location) is exogenous but it is possible that individuals make their housing and work location decisions on the basis of (preferred) travel arrangements. We have however made an attempt to estimate a joint residential location-car ownership-mode choice model for the Dublin city and county sample and obtained results which are consistent with expectations (e.g.,

those living and working near railway stations are significantly more likely to choose all residential location-car ownership-mode options over living outside the city centre, owning a car and travelling by motorised means to work). A further limitation of this analysis concerns the nature of the data available; in particular, information on alternative characteristics such as invehicle time, waiting time, and public transport frequency is simply not available, let alone easily matched to EDs. Our data is also lacking any information on other household members' travel patterns, such as the necessity of dropping children to school, which may influence the car ownership and mode choice decision of an individual. Other potentially important omitted variables include information on the provision of bus services at a detailed regional level and the provision of other services such as cycle lanes, park and ride facilities and dedicated road space for buses.

While necessarily crude, our travel time and cost measures indicate the importance of alternative-specific factors in influencing travel behaviour. A change in the relative time or monetary costs of public transport relative to driving a car are likely to make these more sustainable modes more attractive to commuters. For example, in the Dublin area, increasing the cost of a commuting journey for those who have a car and drive to work by one per cent reduces the probability of that alternative being chosen by approximately 0.05 per cent (and increases the probability of alternatives being chosen by 0.08 per cent), while the respective changes for a similar increase in travel time are approximately 1.2 per cent and 1.7 per cent (see Tables 8 and 9). Faster or cheaper public transport journeys, or relatively more expensive car trips, perhaps through the introduction of road pricing or carbon taxes have the potential to induce modal shifts. The results, while subject to concerns over the reliability of our time and cost measures, suggest that travel time exerts a stronger influence on individual travel behaviour. In this context, measures which seek to make more sustainable modes of transport more comparable with the private car in terms of journey times (e.g., dedicated cycle and bus lanes, more frequent public transport services etc.) may be just as important as monetary incentives in inducing individuals to travel by more sustainable forms of transport.

References

Alperovich, G., Deutsch, J. and Machnes, Y. (1999): 'The Demand for Car Ownership: Evidence from Israeli Data'. *International Journal of Transport Economics*, 26 (3), 351-375.

Asensio, J. (2002): 'Transport Mode Choice by Commuters to Barcelona's CBD'. Urban Studies, 39 (10), 1881-1895.

Barrett, A., Goggin, J. and Kearney, I. (2009). *Quarterly Economic Commentary (Spring 2009)*. Dublin, Economic and Social Research Institute.

Ben-Akiva, M and Lerman, S. (1975): 'Disaggregate Behavioural Model of Automobile Ownership'. *Transportation Research Record*, 569, 34-51.

Bhat, C. and Pulugurtha, V. (1998): 'A Comparison of Two Alternative Behavioural Choice Mechanisms for Household Auto Ownership Decisions'. *Transportation Research Part B: Methodological*, 32(1), 61-75.

Central Statistics Office (2007): Transport 2006. Dublin: Stationery Office.

Central Statistics Office (2008): *Database Direct (Census of Population Interactive Tables)*. Available from: <u>www.cso.ie/px/</u>

Central Statistics Office (2009a). Live Register (April 2009). Dublin: Stationery Office.

Central Statistics Office (2009b). Vehicles Licensed for the First Time (March 2009). Dublin: Stationery Office.

Cohen, A. and Harris, N. (1998): 'Mode Choice for VFR Journeys'. Journal of Transport Geography, 6 (1), 43-51.

Commins, N. and Nolan, A. (2008). *The Determinants of Mode of Transport to Work in the Greater Dublin Area*. ESRI Working Paper No. 268. Dublin: Economic and Social Research Institute.

Cragg, J. and Uhler, R. (1970): 'The Demand for Automobiles'. *Canadian Journal of Economics*, 3 (3), 386-406.

De Donnea, F. (1971): The Determinants of Transport Mode Choice in Dutch Cities: Some Disaggregate Stochastic Choice Models. Rotterdam, Rotterdam University Press.

De Palma, A. and Rochat, D. (2000): 'Mode Choice for Trips to Work in Geneva: An Empirical Analysis'. *Journal of Transport Geography*, 8, 43-51.

Department of Transport (2008a): *Statement of Strategy 2008-2010*. Dublin: Department of Transport.

Department of Transport (2008b): 2020 Vision – Sustainable Transport and Travel. Public Consultation Document. Dublin: Department of Transport.

Dissanayake, D. and Morikawa, T. (2005): 'Household Travel Behavior in Developing Countries: Nested Logit Model of Vehicle Ownership, Mode Choice and Trip Chaining'. *Transportation Research Record*, 1805, 45-52.

Domencich, T. and McFadden, D. (1975): Urban Travel Demand: A Behavioural Analysis. Amsterdam, North Holland.

Dublin Transportation Office (2001): A Platform for Change: An Integrated Transport Strategy for Greater Dublin Area 2000 to 2016. Final Report. Dublin: Dublin Transportation Office.

Dublin Transportation Office (2005): *Road User Monitoring Report 2004*. Dublin: Dublin Transportation Office.

Dublin Transportation Office (2006a): DTO Cycling Policy. Dublin: Dublin Transportation Office.

Dublin Transportation Office (2006b): *Greater Dublin Area Travel Demand Management Study*. Final Report. Prepared by Booz Allen Hamilton Consultants. Dublin: Dublin Transportation Office.

European Commission (2007): *Green Paper: Towards a New Culture for Urban Mobility*. Brussels: Commission of the European Communities.

European Environment Agency (2006): *Urban Sprawl in Europe: the ignored challenge*. EEA Report No 10/2006. Luxembourg: Office for Official Publications of the European Communities.

Ewing, R., Schroeer, W. and Greene, W. (2004): 'School Location and Student Travel: Analysis of Factors Affecting Mode Choice'. *Transportation Research Record*, 1885, 55-53.

Eymann, A. and Ronning, G. (1997): 'Microeconometric Models of Tourists' Destination Choice'. *Regional Science and Urban Economics*, 27, 735-761.

FitzGerald, J., Bergin, A., Conefrey, T., Diffney, S., Duffy, D., Kearney, I., Lyons, S., Malaguzzi Valeri, L., Mayor, K. and Tol, R. (2008): *Medium-Term Review 2008-2015*. Dublin: Economic and Social Research Institute.

Hausman, J. and Wise, D. (1978): 'A Conditional Probit Model for Qualitative Choice: Discrete Decisions Recognizing Interdependence and Heterogeneous Preferences'. *Econometrica*, 46 (2), 403-246.

Hole, A. and FitzRoy, F. (2004): *Commuting in Small Towns in Rural Areas: The Case of St. Andrews.* Department of Economics, University of St. Andrews.

Horner, A. (1999): 'The Tiger stirring: aspects of commuting in the Republic of Ireland 1981–1996'. *Irish Geography*, *32*, *99–111*.

Keane, M. (2001): 'A Model of Commuting Distances: Some Preliminary Insights from a Spatial Model of Job Search'. *European Research in Regional Science*, 11, 169-177.

Lave, C. (1970): 'The Demand for Urban Mass Transportation'. *The Review of Economics and Statistics*, 52 (3), 320-323.

Lave, C. and Train, K. (1979): 'A Disaggregate Model of Auto-Type Choice'. *Transportation Research Part A: Policy and Practice*, 13, 1-9.

Lyons, S., Mayor, K. and Tol, R.S.J. (2008): 'Environmental Accounts of the Republic of Ireland: 1990-2005'. *Journal of the Statistical and Social Inquiry Society of Ireland*, 37, 190-216.

MacLaran, A. and Killen, J., 2002. The Suburbanisation of Office Development in Dublin and its Implications for Transport. *Journal of Irish Urban Studies*, 1 (1), 21-35.

McCafferty, D. and Hall, F. (1982): 'The Use of Multinomial Logit Analysis to Model the Choice of Time to Travel'. *Economic Geography*, *36* (*3*), *236-246*.

McCarthy, C. (2004). Crawling through the Sprawl. Irish Banking Review, Summer, 15-26.

McCarthy, P. (1996): 'Market Price and Income Elasticities of New Vehicle Demands'. *The Review of Economics and Statistics*, 78 (3), 543-547.

McDonnell, S., Ferreira, S. and Convery, F. (2006): 'Impact of Bus Priority Attributes on Catchment Area Residents in Dublin, Ireland'. *Journal of Public Transportation*, 9 (3), 137-162.

McFadden, D. (1974): 'Conditional Logit Analysis of Qualitative Choice Behaviour'. In Zarembka, P. (ed.) *Frontiers in Econometrics*. New York: Academic Press.

McGillivray, R. (1972): 'Binary Choice of Urban Transport Mode in the San Francisco Bay Region'. *Econometrica*, 40 (5), 827-848.

Madan, D. and Groenhout, R. (1987): 'Modelling Travel Mode Choices for the Sydney Work Trip'. *Journal of Transport Economics and Policy*, 21 (2), 135-149.

Morgenroth, E. and FitzGerald, J. (2006): *Ex-Ante Evaluation of the Investment Priorities for the 2007-2013 National Development Plan.* Policy Research Series No. 59. Dublin: Economic and Social Research Institute.

Morgenroth, E. (2002): *Commuting in Ireland: An Analysis of Inter-County Commuting Flows.* ESRI Working Paper No.144. Dublin: Economic and Social Research Institute.

Nolan, A. (2003): 'The Determinants of Urban Households' Transport Decisions: A Microeconometric Study using Irish Data'. *International Journal of Transport Economics*, 30 (1), 103-132.

Pooley, C. and Turnbull, J. (2000): 'Modal Choice and Modal Change: The Journey to Work in Britain since 1890'. *Journal of Transport Geography*, *8*, *11-24*.

Salon, D. (2009). Neighborhoods, cars, and commuting in New York City: A discrete choice approach. *Transportation Research Part A*, 43, 180-196.

Scott Long, J. and Freese, J. (2006): *Regression Models for Categorical Dependent Variables using Stata*. College Station, Texas: Stata Press.

Thobani, M. (1984): 'A Nested Logit Model of Travel Mode to Work and Auto Ownership'. *Journal of Urban Economics*, 15(3), 287-301.

Train, Kenneth (1980): 'A Structured Logit Model of Auto Ownership and Mode Choice'. *The Review of Economic Studies*, 47(2), 357-370.

Vega, A. and Reynolds-Feighan, A. (2006): *Choice of Residential Location and Travel Mode to Work in the Greater Dublin Area.* Paper Presented to UTSG, Dublin, January 2006.

Vega, A. and Reynolds-Feighan, A. (2008): 'Employment Sub-Centres and Travel to Work Mode Choice in the Dublin Region'. *Urban Studies*, 45 (9), 1747-1768.

Walsh, J., Foley, R., Kavanagh, A. and McElwain, A. (2005): *Origins, Destinations and Catchments: Mapping Travel to Work in Ireland in 2002.* Paper presented to the Statistical and Social Inquiry Society of Ireland, October 2005.

Webster, F. and Bly, P. (1980). The Demand for Public Transport: Report of the International Collaborative Study of the Factors Affecting Public Transport Patronage. Crowthorne, Transport and Road Research Laboratory.

Williams, B. and Shiels, P., 2000. Acceleration into Sprawl: Causes and Potential Policy Responses. Article in Quarterly Economic Commentary (Spring 2000). Dublin: Economic and Social Research Institute.

FIGURES AND TABLES





Source: CSO Census Interactive Tables (<u>www.cso.ie</u>)

Table 1: Sub-Sample Definitions and Selected Characteristics

	Dublin city & county	Commuter	Other urban	Rural
Definition	Dublin County Borough, Fingal, South Dublin, Dun Laoghaire-Rathdown	Kildare, Meath and Wicklow	Cork, Galway, Limerick and Waterford cities and EDs with residential density of 150 persons per km ² or greater	EDs with residential density of fewer than 150 persons per km ²
Resident working population	494,370	187,779	377,649	504,532
Population density	4097 ¹⁹	598	1610	46
Average kms to work	10	21	11	18
Median kms to work	7	16	5	12
Public transport options	Extensive bus service; suburban coastal light rail line (DART); four radial suburban heavy rail lines (Commuter); two radial tram lines (LUAS)	Inter-urban bus and rail services; four radial suburban heavy rail lines (Commuter)	City bus services in cities with inter-urban bus and rail services; one suburban rail line in Cork	Inter-urban bus and rail services

The samples exclude those who stated that they work at home, travelled by "other" means (including lorry or van), or did not answer the question (see also Section 4).

Source: 2006 POWCAR

¹⁹ Despite having the highest population density in the country, Dublin is a low density city by European standards (see European Environment Agency, 2006).

	Dublin city & county	Dublin commuter belt	Other Urban	Rural
No household car	14.5	4.7	12.0	2.8
On foot or bicycle	6.9	2.7	8.1	1.7
Bus, train or LUAS	6.8	1.2	2.0	0.3
Motorcycle, scooter, car driver or passenger	0.8	0.8	1.9	0.8
At least one household car	85.5	95.3	88.0	97.2
On foot or bicycle	11.8	7.2	14.0	4.9
Bus, train or LUAS	17.0	9.2	3.2	1.3
Motorcycle, scooter, car driver or passenger	56.7	78.9	70.8	91.0
Total	100.0	100.0	100.0	100.0

Table 2: Household car ownership and mode of transport to work, 2006 (full population of working individuals 15+ years; %)

The samples exclude those who stated that they work at home, travelled by "other" means (including lorry or van), or did not answer the question (see also Section 4). *Source*: 2006 POWCAR

	Definition	Dublin city	Commuter	Other urban	Rural
Age 25-29	=1 if aged 25-29	18.8	15.4	18.4	13.0
Age 30-34	=1 if aged 30-34	16.2	16.8	15.5	15.0
Age 35-39	=1 if aged 35-39	12.2	14.5	12.7	14.8
Age 40-44	=1 if aged 40-44	11.3	12.6	11.4	13.6
Age 45-49	=1 if aged 45-49	10.0	10.8	10.4	12.3
Age 50-54	=1 if aged 50-54	8.5	8.9	8.1	10.1
Age 55-59	=1 if aged 55-59	6.2	5.8	5.7	6.6
Age 60+	=1 if aged 60+ years	4.3	3.7	3.8	4.0
	(Reference category = aged 15-24 years)	12.5	11.5	14.0	10.6
Female	=1 if female	48.5	48.7	49.1	54.2
	(Reference category = male)	51.5	51.3	50.9	45.8
Lone parent with at least	=1 if lone parent with children under 19 years	3.9	3.9	4.8	4.2
Lone parent with resident	=1 if lone parent with children over 19 years	3.6	3.2	3.4	4.1
Couple with at least one resident children under 19	=1 if couple with children under 19 years	33.3	44.1	35.6	49.2
Couple with resident	=1 if couple with children over 19 years	12.0	11.5	10.3	13.2
Couple with no resident children	=1 if couple with no resident children	17.8	19.8	17.8	17.0
Other households	=1 if other household types	20.2	11.0	18.9	6.5
	(Reference category = single households)	9.2	6.5	9.2	5.8
Ever married	=1 if married, separated/divorced, widowed	51.1	62.5	53.9	66.4
	(Reference category = single)	48.9	37.5	46.1	33.6

Table 3: Variable definitions and summa	ry statistics, 2006 (independent variables)
---	---

The samples exclude those who stated that they work at home, travelled by "other" means (including lorry or van), or did not answer the question (see also Section 4).

Table	3:	continue	d
	_		

		Dublin city	Commuter	Other	Rural
		& county		urban	
Third level	=1 if highest level of education completed is third level	54.5	47.0	46.0	42.0
	(Reference category = less than third level)	45.5	53.0	54.0	58.0
Employers or managers	=1 if employer or manager	20.7	20.2	14.7	14.3
Higher professional	=1 if higher professional	12.3	7.6	8.7	6.3
Lower professional	=1 if lower professional	17.1	16.6	16.2	17.9
-	(Reference category = all other socio-economic groups)				
Commerce	=1 if works in commerce	41.1	33.0	30.1	26.8
Public administration	=1 if works in public administration or defence	7.9	7.8	6.2	6.7
Health, education, social	=1 if works in health, education or social work	19.2	19.5	21.5	25.7
	(Reference category = all other industrial groups)	49.9	55.6	60.4	61.5
Population density (home)	=1 if population density of home ED is $>=150$ per km ²	98.4	63.0	*	*
Population density (work)	=1 if population density of work ED is $>=150$ per km ²	97.5	80.5	83.6	53.9
Rail available	=1 if lives and works in an ED where 100 per cent of	34.8	4.1		
	addresses are within 2 kilometres of a rail station (Reference category = does not live and work in such an ED)	65.2	95.9		
Rail available	=1 if lives and works in an ED where 75 per cent of addresses are within 2 kilometres of a rail station (Reference category = does not live and work in such an ED)			19.6 80.4	2.1 97.9

The samples exclude those who stated that they work at home, travelled by "other" means (including lorry or van), or did not answer the question (see also Section 4).

* As the urban and rural samples are defined on the basis of population density above and below 150 persons per km^2 (see Section 3), the population density (home) variable drops out of the analysis in these two sub-samples.

	No car		•	One or r	nore cars
	On foot or bicycle	Bus or train	Motorcycle, car driver, car passenger	On foot or bicycle	Bus or train
Individual-specific variables					
Age 15-24	ref	ref	ref	ref	ref
Age 25-29	0.77***	0.68***	0.62**	0.69***	0.60***
Age 30-34	0.48^{***}	0.42***	0.70*	0.68***	0.48***
Age 35-39	0.41***	0.40***	0.62*	0.57***	0.39***
Age 40-44	0.33***	0.31***	0.46***	0.58***	0.39***
Age 45-49	0.34***	0.36***	0.35***	0.61***	0.36***
Age 50-54	0.25***	0.29***	0.31***	0.52***	0.34***
Age 55-59	0.19***	0.28***	0.56*	0.49***	0.31***
Age 60+	0.16***	0.22***	0.30***	0.35***	0.35***
Male	ref	ref	ref	ref	ref
Female	0.66***	0.88**	0.56***	0.79***	1.33***
Single Person	ref	ref	ref	ref	ref
Lone parent with at least one resident child under 19	0.37***	0.37***	0.46**	0.87	1.15
Lone parent with resident children but none under 19	0.31***	0.51***	0.65*	1.11	1.94***
Couple with at least one resident children under 19	0.09***	0.11***	0.13***	1.19**	1.47***
Couple with resident children but none under 19	0.11***	0.11***	0.16***	1.38***	2.17***
Couple with no resident children	0.50***	0.44***	0.23***	1.47***	1.81***
Other households	0.88	0.99	0.58***	2.40***	2.17***
Single	ref	ref	ref	ref	ref
Ever married	0.83**	0.79***	0.66**	0.78	0.76***
Less than third level	ref	ref	ref	ref	ref
Third level	0.91	0.72***	0.51***	1.16***	1.19***

Table 4: Dublin City and County (Odds Ratios – Reference Choice is Car Owner and Motorcycle, Car Driver or Car Passenger)

Table 4: continued

	No car			One or m	ore cars
	On foot or	Bus or train	Motorcycle,	On foot or	Bus or train
	bicycle		car driver, car	bicycle	
			passenger		
Employers and managers	0.26***	0.37***	0.26***	0.57***	0.66***
Higher professional	0.47***	0.37***	0.32***	1.06	0.78***
Lower professional	0.49***	0.54***	0.45***	0.89*	0.89**
All other socio-economic groups	ref	ref	ref	ref	ref
Commerce	1.09	1.12**	0.98	1.49***	1.83***
Public administration and defence	1.24*	1.11	0.49**	1.92***	2.19***
Education, health and social work	0.90	0.75***	0.80	1.04	0.73***
All other industrial groups	ref	ref	ref	ref	ref
Population density (home)	2.26*	3.78***	1.24	1.28	1.68***
Population density (work)	1.22	1.51**	0.78	1.66***	3.23***
Living and working in an ED with less than 100 per cent of addresses within 2kms of a rail station Living and working in an ED with 100 per cent of	ref	ref	ref	ref	ref
addresses within 2kms of a rail station	3.62***	2.72***	2.01***	2.32***	1.88***
Alternative-specific variables					
Travel time			0.77***		
Travel cost			0.98***		
Number of Observations			213,168		
Number of Individuals			35,528		
Log-Likelihood			-34,877.82		

	No car			One or n	nore cars
	On foot or bicycle	Bus or train	Motorcycle, car driver, car passenger	On foot or bicycle	Bus or train
Individual-specific variables					
Age 15-24	ref	ref	ref	ref	ref
Age 25-29	0.90	0.81	1.08	0.66***	0.58***
Age 30-34	0.75	0.67	0.92	0.41***	0.48***
Age 35-39	0.61	0.33***	0.68	0.53***	0.43***
Age 40-44	1.01	0.42**	0.93	0.65**	0.43***
Age 45-49	0.71	0.90	0.33*	0.54***	0.45***
Age 50-54	0.36***	0.47*	0.51	0.49***	0.47***
Age 55-59	0.90	0.80	0.41	0.58**	0.39***
Age 60+	0.53	0.19**	0.34	0.68	0.42***
Male	ref	ref	ref	ref	ref
Female	0.74**	0.82	0.33***	1.05	1.28***
Single Person	ref	ref	ref	ref	ref
Lone parent with at least one resident child under 19	0.26	0.45**	2.83**	0.86	1.15
Lone parent with resident children but none under 19	0.43**	0.45**	0.80	2.20**	2.25***
Couple with at least one resident children under 19	0.05***	0.05***	0.17***	1.78**	1.48**
Couple with resident children but none under 19	0.06***	0.07***	0.16***	1.99**	2.02***
Couple with no resident children	0.17***	0.25***	0.23***	2.01**	1.57***
Other households	0.94	0.59**	1.61	4.98***	1.97***
Single	ref	ref	ref	ref	ref
Ever married	0.91***	0.67*	0.52**	0.97	0.79**
Less than third level	ref	ref	ref	ref	ref
Third level	0.54**	0.85	0.43***	0.84*	1.50***

Table 5: Commuter Counties - Kildare, Meath, Wicklow (Odds Ratios – Reference Choice is Car Owner and Motorcycle, Car Driver or Car Passenger)

Table 5 continued

	No car			One or more cars		
	On foot or	Bus or train	Motorcycle,	On foot or	Bus or train	
	bicycle		car driver, car	bicycle		
			passenger			
Employers and managers	0.37***	0.43***	0.23***	0.52***	1.04	
Higher professional	0.27**	0.72	0.65	0.89	1.18	
Lower professional	0.19***	0.73	0.55	0.51***	0.99	
All other socio-economic groups	ref	ref	ref	ref	ref	
Commerce	1.04	1.54**	0.69	1.24**	2.16***	
Public administration and defence	0.80	1.05	0.45	0.88	3.51***	
Education, health and social work	1.17	0.64	0.62	0.98	1.05	
All other industrial groups	ref	ref	ref	ref	ref	
Population density (home)	1.18	4.07***	1.50*	1.38***	2.24****	
Population density (work)	1.99***	2.12***	0.59**	1.22*	5.01***	
Living and working in an ED with less than 100 per cent of addresses within 2kms of a rail station	ref	ref	ref	ref	ref	
addresses within 2kms of a rail station	3.38***	1.39	1.44	2.03***	2.47***	
Alternative-specific variables						
Travel time			0.75***			
Travel cost			1.00			
Number of Observations			83,376			
Number of Individuals			13,896			
Log-Likelihood			-7,219.9			

	No car			One or more cars	
	On foot or bicycle	Bus or train	Motorcycle, car driver, car passenger	On foot or bicycle	Bus or train
Individual-specific variables					
Age 15-24	ref	ref	ref	ref	ref
Age 25-29	0.70***	0.72**	0.88	0.66***	0.58***
Age 30-34	0.44***	0.64***	0.76*	0.52***	0.51***
Age 35-39	0.37***	0.42***	0.82	0.53***	0.60***
Age 40-44	0.37***	0.45***	0.71*	0.63***	0.52***
Age 45-49	0.41***	0.52***	0.51***	0.63***	0.46***
Age 50-54	0.50***	0.29***	0.68	0.61***	0.38***
Age 55-59	0.30***	0.65*	0.54**	0.55***	0.38***
Age 60+	0.21***	0.38***	0.56*	0.46***	0.32***
Male	ref	ref	ref	ref	ref
Female	0.66***	0.88	0.75***	1.04	1.08
Single Person	ref	ref	ref	ref	ref
Lone parent with at least one resident child under 19	0.48***	0.44***	0.77	0.78*	1.19
Lone parent with resident children but none under 19	0.52***	0.53***	0.78	1.80***	1.91***
Couple with at least one resident children under 19	0.09***	0.09***	0.22***	1.14	1.22
Couple with resident children but none under 19	0.08***	0.12***	0.11***	1.47***	2.11***
Couple with no resident children	0.28***	0.31***	0.26***	1.38***	1.74***
Other households	0.96	1.12	1.13	3.06***	2.50***
Single	ref	ref	ref	ref	ref
Ever married	1.06	0.88	0.81	0.93	0.76**
Less than third level	ref	ref	ref	ref	ref
Third level	0.68***	0.79**	0.51***	0.86***	1.10

 Table 6: Other Urban Areas (Odds Ratios – Reference Choice is Car Owner and Motorcycle, Car Driver or Car Passenger)

Table 6: continued

	No car			One or 1	One or more cars		
	On foot or bicycle	Bus or train	Motorcycle, car driver, car	On foot or bicycle	Bus or train		
			passenger				
Employers and managers	0.30***	0.33***	0.29***	0.58***	0.61***		
Higher professional	0.20***	0.24***	0.17***	0.64^{***}	0.85		
Lower professional	0.32***	0.44 * * *	0.37***	0.52***	0.81*		
All other socio-economic groups	ref	ref	ref	ref	ref		
Commerce	0.91	0.79**	0.77**	1.12**	1.47***		
Public administration and defence	0.43***	0.56**	0.36***	0.69***	1.24		
Education, health and social work	0.92	0.46***	0.43***	0.96	0.69***		
All other industrial groups	ref	ref	ref	ref	ref		
Population density (work)	1.07	1.64***	0.64***	1.91***	1.94***		
Living and working in an ED with less than 75 per cent of addresses within 2kms of a rail station Living and working in an ED with greater than 75 per	ref	ref	ref	ref	ref		
cent of addresses within 2kms of a rail station	2.25***	0.62***	1.11	1.72***	0.86		
Alternative-specific variables							
Travel time			0.83***				
Travel cost			0.99***				
Number of Observations			161,394				
Number of Individuals			26,899				
Log-Likelihood			-18,554.58				

	No car			One or more cars	
	On foot or bicycle	Bus or train	Motorcycle, car driver, car passenger	On foot or bicycle	Bus or train
Individual-specific variables					
Age 15-24	ref	ref	ref	ref	ref
Age 25-29	1.06	1.52	0.83	0.73**	0.53***
Age 30-34	0.70*	0.78	0.92	0.55***	0.48***
Age 35-39	0.56**	0.32*	0.62*	0.66***	0.52***
Age 40-44	0.77	0.74	0.60*	0.61***	0.55***
Age 45-49	0.61**	0.82	0.85	0.73**	0.43***
Age 50-54	0.52**	1.07	0.82	0.62***	0.60**
Age 55-59	0.56**	0.51	0.91	0.62***	0.75
Age 60+	0.26***	0.52	0.50*	0.60***	0.68
Male	ref	ref	ref	ref	ref
Female	0.92	0.57**	0.69**	0.97	1.30***
Single Person	ref	ref	ref	ref	ref
Lone parent with at least one resident child under 19	0.43***	0.21**	0.72	0.63**	2.90***
Lone parent with resident children but none under 19	0.34***	0.45*	0.54***	1.09	3.63***
Couple with at least one resident children under 19	0.06***	0.02***	0.06***	0.90	3.10***
Couple with resident children but none under 19	0.08***	0.07***	0.07***	1.01	3.11***
Couple with no resident children	0.22***	0.07**	0.10***	1.02	3.54***
Other households	1.13	1.14	0.80	2.12***	4.33***
Single	ref	ref	ref	ref	ref
Ever married	0.97	0.95	0.81	0.81**	0.47***
Less than third level	ref	ref	ref	ref	ref
Third level	0.59***	0.50**	0.39***	0.99	0.81*

Table 7: Rural Areas (Odds Ratios – Reference Choice is Car Owner and Motorcycle, Car Driver or Car Passenger)

Table 7: continued

		No car		One or 1	nore cars
	On foot or bicycle	Bus or train	Motorcycle, car driver, car passenger	On foot or bicycle	Bus or train
Employers and managers	0.28***	0.39*	0.24***	0.79**	0.93
Higher professional	0.18***	1.00	0.18**	0.47***	0.95
Lower professional	0.25***	0.18*	0.58*	0.46***	0.62***
All other socio-economic groups	ref	ref	ref	ref	ref
Commerce	0.59***	0.95	0.71**	0.95	1.35***
Public administration and defence	0.30**	1.18	0.36**	0.77	2.18***
Education, health and social work	0.68**	1.34	0.75	0.85	1.06
All other industrial groups	ref	ref	ref	ref	ref
Population density (work)	0.66***	1.31	0.45***	0.63***	2.56***
Living and working in an ED with less than 75 per cent of addresses within 2kms of a rail station Living and working in an ED with greater than 75 per	ref	ref	ref	ref	ref
cent of addresses within 2kms of a rail station	3.10***	3.12**	1.16	1.21	1.75**
Alternative-specific variables					
Travel time			0.75***		
Travel cost			0.96***		
Number of Observations			211,752		
Number of Individuals			35,292		
Log-Likelihood			-8,592.00		

	No car- walk/cycle	No car-public transport	No car-motor	Car-walk/cycle	Car-public transport	Car-motor
	<u> </u>	1	Dublin cit	ty & county	L	
No car-walk/cycle	-7.1	0.2	0.0	0.7	0.5	1.7
No car-public transport	0.4	-3.0	0.0	0.7	0.5	1.7
No car-motor	0.4	0.2	-2.6	0.7	0.5	1.7
Car-walk/cycle	0.4	0.2	0.0	-5.7	0.5	1.7
Car-public transport	0.4	0.2	0.0	0.7	-2.7	1.7
Car-motor	0.4	0.2	0.0	0.7	0.5	-1.2
			Com	muter		
No car-walk/cycle	-21.2	0.1	0.0	0.9	0.6	5.2
No car-public transport	0.5	-7.7	0.0	0.9	0.6	5.2
No car-motor	0.5	0.1	-6.2	0.9	0.6	5.2
Car-walk/cycle	0.5	0.1	0.0	-14.3	0.6	5.2
Car-public transport	0.5	0.1	0.0	0.9	-5.4	5.2
Car-motor	0.5	0.1	0.0	0.9	0.6	-1.3
			Other	r urban		
No car-walk/cycle	-7.7	0.0	0.0	1.0	0.1	1.4
No car-public transport	0.5	-1.9	0.0	1.0	0.1	1.4
No car-motor	0.5	0.0	-1.7	1.0	0.1	1.4
Car-walk/cycle	0.5	0.0	0.0	-6.8	0.1	1.4
Car-public transport	0.5	0.0	0.0	1.0	-2.0	1.4
Car-motor	0.5	0.0	0.0	1.0	0.1	-0.5
			R	ural		
No car-walk/cycle	-18.8	0.0	0.0	0.6	0.1	4.3
No car-public transport	0.2	-6.0	0.0	0.6	0.1	4.3
No car-motor	0.2	0.0	-4.7	0.6	0.1	4.3
Car-walk/cycle	0.2	0.0	0.0	-15.4	0.1	4.3
Car-public transport	0.2	0.0	0.0	0.6	-4.8	4.3
Car-motor	0.2	0.0	0.0	0.6	0.1	-0.4

Table 8: Travel Time Elasticities

See http://pages.stern.nyu.edu/~wgreene/BergenMasterclass/DiscreteChoiceLab.ppt#766,117,Effects of Changes in Attributes on Probabilities for further details. Cross elasticities are identical across alternatives due to the IIA assumption.

	No car- walk/cycle	No car-public transport	No car-motor	Car-walk/cycle	Car-public transport	Car-motor
		1	Dublin cit	ty & county	1	
No car-walk/cycle	0.0	0.0	0.0	0.0	0.0	0.1
No car-public transport	0.0	-0.1	0.0	0.0	0.0	0.1
No car-motor	0.0	0.0	0.0	0.0	0.0	0.1
Car-walk/cycle	0.0	0.0	0.0	0.0	0.0	0.1
Car-public transport	0.0	0.0	0.0	0.0	-0.1	0.1
Car-motor	0.0	0.0	0.0	0.0	0.0	-0.1
			Com	nmuter		
No car-walk/cycle	0.0	0.0	0.0	0.0	0.0	0.0
No car-public transport	0.0	0.0	0.0	0.0	0.0	0.0
No car-motor	0.0	0.0	0.0	0.0	0.0	0.0
Car-walk/cycle	0.0	0.0	0.0	0.0	0.0	0.0
Car-public transport	0.0	0.0	0.0	0.0	0.0	0.0
Car-motor	0.0	0.0	0.0	0.0	0.0	0.0
			Other	r urban		
No car-walk/cycle	0.0	0.0	0.0	0.0	0.0	0.0
No car-public transport	0.0	0.0	0.0	0.0	0.0	0.0
No car-motor	0.0	0.0	0.0	0.0	0.0	0.0
Car-walk/cycle	0.0	0.0	0.0	0.0	0.0	0.0
Car-public transport	0.0	0.0	0.0	0.0	0.0	0.0
Car-motor	0.0	0.0	0.0	0.0	0.0	0.0
			R	ural		
No car-walk/cycle	0.0	0.0	0.0	0.0	0.0	0.4
No car-public transport	0.0	-0.1	0.0	0.0	0.0	0.4
No car-motor	0.0	0.0	0.0	0.0	0.0	0.4
Car-walk/cycle	0.0	0.0	0.0	-0.2	0.0	0.4
Car-public transport	0.0	0.0	0.0	0.0	-0.3	0.4
Car-motor	0.0	0.0	0.0	0.0	0.0	0.0

Table 9: Travel Cost Elasticities

See http://pages.stern.nyu.edu/~wgreene/BergenMasterclass/DiscreteChoiceLab.ppt#766,117,Effects of Changes in Attributes on Probabilities for further details. Cross elasticities are identical across alternatives due to the IIA assumption.

Mara	N 1	Title/Author(s)
Year	Number	ESRI Authors/Co-authors Italicised
2009	309	Recent Trends in the Caesarean Section Rate in Ireland 1999-2006 <i>Aoife Brick</i> and <i>Richard Layte</i>
	308	Price Inflation and Income Distribution Anne Jennings, Seán Lyons and Richard S.J. Tol
	307	Overskilling Dynamics and Education Pathways Kostas Mavromaras, <i>Seamus McGuinness</i> , Yin King Fok
	306	What Determines the Attractiveness of the European Union to the Location of R&D Multinational Firms? <i>Iulia Siedschlag, Donal Smith, Camelia Turcu, Xiaoheng Zhang</i>
	305	Do Foreign Mergers and Acquisitions Boost Firm Productivity? <i>Marc Schiffbauer, Iulia Siedschlag, Frances Ruane</i>
	304	Inclusion or Diversion in Higher Education in the Republic of Ireland? <i>Delma Byrne</i>
	303	Welfare Regime and Social Class Variation in Poverty and Economic Vulnerability in Europe: An Analysis of EU-SILC Christopher T. Whelan and <i>Bertrand Maître</i>
	302	Understanding the Socio-Economic Distribution and Consequences of Patterns of Multiple Deprivation: An Application of Self-Organising Maps Christopher T. Whelan, Mario Lucchini, Maurizio Pisati and <i>Bertrand Maître</i>
	301	Estimating the Impact of Metro North Edgar Morgenroth
	300	Explaining Structural Change in Cardiovascular Mortality in Ireland 1995-2005: A Time Series Analysis <i>Richard Layte, Sinead O'Hara</i> and Kathleen Bennett
	299	EU Climate Change Policy 2013-2020: Using the Clean Development Mechanism More Effectively <i>Paul K Gorecki, Seán Lyons</i> and <i>Richard S.J. Tol</i>
	298	Irish Public Capital Spending in a Recession Edgar Morgenroth

297	Exporting and Ownership Contributions to Irish Manufacturing Productivity Growth Anne Marie Gleeson, <i>Frances Ruane</i>
296	Eligibility for Free Primary Care and Avoidable Hospitalisations in Ireland <i>Anne Nolan</i>
295	Managing Household Waste in Ireland: Behavioural Parameters and Policy Options John Curtis, Seán Lyons and Abigail O'Callaghan-Platt
294	Labour Market Mismatch Among UK Graduates; An Analysis Using REFLEX Data <i>Seamus McGuinness</i> and <i>Peter J. Sloane</i>
293	Towards Regional Environmental Accounts for Ireland <i>Richard S.J. Tol , Nicola Commins, Niamh Crilly, Sean Lyons</i> and <i>Edgar Morgenroth</i>
292	EU Climate Change Policy 2013-2020: Thoughts on Property Rights and Market Choices <i>Paul K. Gorecki, Sean Lyons</i> and <i>Richard S.J. Tol</i>
291	Measuring House Price Change David Duffy
290	Intra-and Extra-Union Flexibility in Meeting the European Union's Emission Reduction Targets <i>Richard S.J. Tol</i>
289	The Determinants and Effects of Training at Work: Bringing the Workplace Back In <i>Philip J. O'Connell</i> and <i>Delma Byrne</i>
288	Climate Feedbacks on the Terrestrial Biosphere and the Economics of Climate Policy: An Application of <i>FUND Richard S.J. Tol</i>
287	The Behaviour of the Irish Economy: Insights from the HERMES macro-economic model Adele Bergin, Thomas Conefrey, John FitzGerald and Ide Kearney
286	Mapping Patterns of Multiple Deprivation Using Self-Organising Maps: An Application to EU-SILC Data for Ireland Maurizio Pisati, <i>Christopher T. Whelan</i> , Mario Lucchini and <i>Bertrand Maître</i>

285	The Feasibility of Low Concentration Targets: An Application of FUND <i>Richard S.J. Tol</i>
284	Policy Options to Reduce Ireland's GHG Emissions
	Instrument choice: the pros and cons of alternative policy instruments Thomas Legge and <i>Sue Scott</i>
283	Accounting for Taste: An Examination of Socioeconomic Gradients in Attendance at Arts Events <i>Pete Lunn</i> and <i>Elish Kelly</i>
282	The Economic Impact of Ocean Acidification on Coral Reefs Luke M. Brander, Katrin Rehdanz, <i>Richard S.J. Tol</i> , and Pieter J.H. van Beukering
281	Assessing the impact of biodiversity on tourism flows: A model for tourist behaviour and its policy implications Giulia Macagno, Maria Loureiro, Paulo A.L.D. Nunes and <i>Richard S.J. Tol</i>
280	Advertising to boost energy efficiency: the Power of One campaign and natural gas consumption <i>Seán Diffney, Seán Lyons</i> and <i>Laura Malaguzzi Valeri</i>
279	International Transmission of Business Cycles Between Ireland and its Trading Partners Jean Goggin and Iulia Siedschlag
278	Optimal Global Dynamic Carbon Taxation David Anthoff
277	Energy Use and Appliance Ownership in Ireland <i>Eimear Leahy</i> and <i>Seán Lyons</i>
276	Discounting for Climate Change David Anthoff, Richard S.J. Tol and Gary W. Yohe
275	Projecting the Future Numbers of Migrant Workers in the Health and Social Care Sectors in Ireland <i>Alan Barrett</i> and Anna Rust
274	Economic Costs of Extratropical Storms under Climate Change: An application of FUND Daiju Narita, <i>Richard S.J. Tol, David Anthoff</i>
273	The Macro-Economic Impact of Changing the Rate of Corporation Tax <i>Thomas Conefrey</i> and <i>John D. Fitz Gerald</i>

272	The Games We Used to Play An Application of Survival Analysis to the Sporting Life- course <i>Pete Lunn</i>
271	Exploring the Economic Geography of Ireland Edgar Morgenroth
270	Benchmarking, Social Partnership and Higher Remuneration: Wage Settling Institutions and the Public-Private Sector Wage Gap in Ireland Elish Kelly, Seamus McGuinness, Philip O'Connell
269	A Dynamic Analysis of Household Car Ownership in Ireland <i>Anne Nolan</i>
268	The Determinants of Mode of Transport to Work in the Greater Dublin Area <i>Nicola Commins</i> and <i>Anne Nolan</i>
267	Resonances from <i>Economic Development</i> for Current Economic Policymaking <i>Frances Ruane</i>
266	The Impact of Wage Bargaining Regime on Firm-Level Competitiveness and Wage Inequality: The Case of Ireland Seamus McGuinness, Elish Kelly and Philip O'Connell
265	Poverty in Ireland in Comparative European Perspective <i>Christopher T. Whelan</i> and <i>Bertrand Maître</i>
264	A Hedonic Analysis of the Value of Rail Transport in the Greater Dublin Area <i>Karen Mayor, Seán Lyons, David Duffy</i> and <i>Richard S.J.</i> <i>Tol</i>
263	Comparing Poverty Indicators in an Enlarged EU Christopher T. Whelan and Bertrand Maître
262	Fuel Poverty in Ireland: Extent, Affected Groups and Policy Issues <i>Sue Scott, Seán Lyons, Claire Keane,</i> Donal McCarthy and <i>Richard S.J. Tol</i>
261	The Misperception of Inflation by Irish Consumers <i>David Duffy</i> and <i>Pete Lunn</i>
260	The Direct Impact of Climate Change on Regional

	Labour Productivity Tord Kjellstrom, R Sari Kovats, Simon J. Lloyd, Tom Holt, <i>Richard S.J. Tol</i>
259	Damage Costs of Climate Change through Intensification of Tropical Cyclone Activities: An Application of FUND Daiju Narita, <i>Richard S. J. Tol</i> and <i>David Anthoff</i>
258	Are Over-educated People Insiders or Outsiders? A Case of Job Search Methods and Over-education in UK Aleksander Kucel, <i>Delma Byrne</i>
257	Metrics for Aggregating the Climate Effect of Different Emissions: A Unifying Framework <i>Richard S.J. Tol,</i> Terje K. Berntsen, Brian C. O'Neill, Jan S. Fuglestvedt, Keith P. Shine, Yves Balkanski and Laszlo Makra
256	Intra-Union Flexibility of Non-ETS Emission Reduction Obligations in the European Union <i>Richard S.J. Tol</i>
255	The Economic Impact of Climate Change Richard S.J. Tol
254	Measuring International Inequity Aversion <i>Richard S.J. Tol</i>
253	Using a Census to Assess the Reliability of a National Household Survey for Migration Research: The Case of Ireland <i>Alan Barrett</i> and <i>Elish Kelly</i>
252	Risk Aversion, Time Preference, and the Social Cost of Carbon David Anthoff, Richard S.J. Tol and Gary W. Yohe
251	The Impact of a Carbon Tax on Economic Growth and Carbon Dioxide Emissions in Ireland <i>Thomas Conefrey, John D. Fitz Gerald, Laura Malaguzzi</i> <i>Valeri</i> and <i>Richard S.J. Tol</i>
250	The Distributional Implications of a Carbon Tax in Ireland <i>Tim Callan, Sean Lyons, Susan Scott, Richard S.J. Tol</i> and Stefano Verde
249	Measuring Material Deprivation in the Enlarged EU <i>Christopher T. Whelan, Brian Nolan</i> and <i>Bertrand Maître</i>
248	Marginal Abatement Costs on Carbon-Dioxide Emissions:

	A Meta-Analysis Onno Kuik, Luke Brander and <i>Richard S.J. Tol</i>
247	Incorporating GHG Emission Costs in the Economic Appraisal of Projects Supported by State Development Agencies <i>Richard S.J. Tol</i> and <i>Seán Lyons</i>
246	A Carton Tax for Ireland Richard S.J. Tol, Tim Callan, Thomas Conefrey, John D. Fitz Gerald, Seán Lyons, Laura Malaguzzi Valeri and Susan Scott
245	Non-cash Benefits and the Distribution of Economic Welfare Tim Callan and Claire Keane
244	Scenarios of Carbon Dioxide Emissions from Aviation Karen Mayor and Richard S.J. Tol
243	The Effect of the Euro on Export Patterns: Empirical Evidence from Industry Data <i>Gavin Murphy</i> and <i>Iulia Siedschlag</i>
242	The Economic Returns to Field of Study and Competencies Among Higher Education Graduates in Ireland <i>Elish Kelly, Philip O'Connell</i> and <i>Emer Smyth</i>
241	European Climate Policy and Aviation Emissions Karen Mayor and Richard S.J. Tol
240	Aviation and the Environment in the Context of the EU- US Open Skies Agreement <i>Karen Mayor</i> and <i>Richard S.J. Tol</i>
239	Yuppie Kvetch? Work-life Conflict and Social Class in Western Europe <i>Frances McGinnity</i> and <i>Emma Calvert</i>
238	Immigrants and Welfare Programmes: Exploring the Interactions between Immigrant Characteristics, Immigrant Welfare Dependence and Welfare Policy <i>Alan Barrett</i> and Yvonne McCarthy
237	How Local is Hospital Treatment? An Exploratory Analysis of Public/Private Variation in Location of Treatment in Irish Acute Public Hospitals Jacqueline O'Reilly and Miriam M. Wiley
236	The Immigrant Earnings Disadvantage Across the Earnings and Skills Distributions: The Case of Immigrants from the EU's New Member States in Ireland

	Alan Barrett, Seamus McGuinness and Martin O'Brien
235	Europeanisation of Inequality and European Reference Groups <i>Christopher T. Whelan</i> and <i>Bertrand Maître</i>
234	Managing Capital Flows: Experiences from Central and Eastern Europe Jürgen von Hagen and <i>Iulia Siedschlag</i>
233	ICT Diffusion, Innovation Systems, Globalisation and Regional Economic Dynamics: Theory and Empirical Evidence Charlie Karlsson, Gunther Maier, Michaela Trippl, <i>Iulia</i> <i>Siedschlag,</i> Robert Owen and <i>Gavin Murphy</i>
232	Welfare and Competition Effects of Electricity Interconnection between Great Britain and Ireland Laura Malaguzzi Valeri
231	Is FDI into China Crowding Out the FDI into the European Union? Laura Resmini and <i>Iulia Siedschlag</i>
230	Estimating the Economic Cost of Disability in Ireland John Cullinan, Brenda Gannon and Seán Lyons
229	Controlling the Cost of Controlling the Climate: The Irish Government's Climate Change Strategy Colm McCarthy, <i>Sue Scott</i>
228	The Impact of Climate Change on the Balanced-Growth- Equivalent: An Application of <i>FUND</i> <i>David Anthoff, Richard S.J. Tol</i>
227	Changing Returns to Education During a Boom? The Case of Ireland Seamus McGuinness, Frances McGinnity, Philip O'Connell
226	'New' and 'Old' Social Risks: Life Cycle and Social Class Perspectives on Social Exclusion in Ireland <i>Christopher T. Whelan</i> and <i>Bertrand Maître</i>
225	The Climate Preferences of Irish Tourists by Purpose of Travel <i>Seán Lyons, Karen Mayor</i> and <i>Richard S.J. Tol</i>
224	A Hirsch Measure for the Quality of Research Supervision, and an Illustration with Trade Economists <i>Frances P. Ruane</i> and <i>Richard S.J. Tol</i>
223	Environmental Accounts for the Republic of Ireland:

		1990-2005 <i>Seán Lyons, Karen Mayor</i> and <i>Richard S.J. Tol</i>
2007	222	Assessing Vulnerability of Selected Sectors under Environmental Tax Reform: The issue of pricing power <i>J. Fitz Gerald</i> , M. Keeney and <i>S. Scott</i>
	221	Climate Policy Versus Development Aid Richard S.J. Tol
	220	Exports and Productivity – Comparable Evidence for 14 Countries <i>The International Study Group on Exports and</i> <i>Productivity</i>
	219	Energy-Using Appliances and Energy-Saving Features: Determinants of Ownership in Ireland Joe O'Doherty, <i>Seán Lyons</i> and <i>Richard S.J. Tol</i>
	218	The Public/Private Mix in Irish Acute Public Hospitals: Trends and Implications Jacqueline O'Reilly and Miriam M. Wiley
	217	Regret About the Timing of First Sexual Intercourse: The Role of Age and Context <i>Richard Layte</i> , Hannah McGee
	216	Determinants of Water Connection Type and Ownership of Water-Using Appliances in Ireland Joe O'Doherty, <i>Seán Lyons</i> and <i>Richard S.J. Tol</i>
	215	Unemployment – Stage or Stigma? Being Unemployed During an Economic Boom <i>Emer Smyth</i>
	214	The Value of Lost Load Richard S.J. Tol
	213	Adolescents' Educational Attainment and School Experiences in Contemporary Ireland <i>Merike Darmody, Selina McCoy, Emer Smyth</i>
	212	Acting Up or Opting Out? Truancy in Irish Secondary Schools <i>Merike Darmody, Emer Smyth</i> and <i>Selina McCoy</i>
	211	Where do MNEs Expand Production: Location Choices of the Pharmaceutical Industry in Europe after 1992 <i>Frances P. Ruane</i> , Xiaoheng Zhang
	210	Holiday Destinations: Understanding the Travel Choices of Irish Tourists

	Seán Lyons, Karen Mayor and Richard S.J. Tol
209	The Effectiveness of Competition Policy and the Price- Cost Margin: Evidence from Panel Data Patrick McCloughan, <i>Seán Lyons</i> and William Batt
208	Tax Structure and Female Labour Market Participation: Evidence from Ireland <i>Tim Callan</i> , A. Van Soest, <i>J.R. Walsh</i>
207	Distributional Effects of Public Education Transfers in Seven European Countries <i>Tim Callan,</i> Tim Smeeding and Panos Tsakloglou